Author's response to reviews

Title: EUS assisted transmural cholecystogastrostomy fistula creation as a bridge for endoscopic internal gallbladder therapy using a novel fully covered metal stent

Authors:

Nan Ge (ge_nan2013@163.com)
zhiguo Wang (wangzq041510102@163.com)
Siyu Sun (sun-siyu@163.com)
Sheng Wang (wangs@sj-hospital.org)
Guoxin Wang (wanggx@sj-hospital.org)
Shiwei Sun (sunsw@sj-hospital.org)
Linlin Feng (1987a67@163.com)
Fei yang (lakeblue18@163.com)
Wenzhuang Ma (mawenzhuang040505@126.com)
Shupeng Wang (wangshupengyd@126.com)
Xiang Liu (Liux@sj-hospital.org)
Jintao Guo (guoit@sj-hospital.org)
Wen Liu (Lavinia1019@163.com)

Version: 5 Date: 6 September 2014

Author's response to reviews: see over
Dear editor,

I would thank you for considering this paper publishing in your journal and also thank for the reviewers’ supports. The manuscript has been modified according to the suggestion and submitted to the tracking system.

The following is a point-to-point response to the reviewers' comments.

ARRIVE
The AARIVE checklist has completed and uploaded to the journal.

2. Figures
According to the journal guidelines, the figures have changed and rearranged into PPTx form.

3. Please format the Title page
The authors’ email addresses have added in the title page.

4. Please format your manuscript following the journal style
The format has changed according the journal’s guideline.

5. Keywords
The Keywords have added after the abstract.

6. Please provide Background section heading.

7. Please format Author’s contribution section.
The Author’s contribution has been rewrite.

8. Request for acknowledgment section.
The acknowledgement section containing the supporting fund has been added after the Author's contribution section.